Author’s response to reviews

Title: Jingfukang induces anti-cancer activity through oxidative stress-mediated DNA damage in circulating human lung cancer cells

Authors:

Zujun Que (030110371@mail.ecust.edu.cn)
Zhiyi Zhou (zhouzhiyi77@hotmail.com)
Bin Luo (luo2014bin@163.com)
Changsheng Dong (dongchangsheng2010@163.com)
Yi Jiang (877708348@qq.com)
Hegen Li (shlaogen@163.com)
Jianhui Tian (hawk7150@hotmail.com)

Version: 3 Date: 26 Sep 2018

Author’s response to reviews:

List of Responses

Dear Editor and Reviewers:

Thank you very much for your kindness and for the reviewers’ comments concerning our manuscript entitled “Jingfukang induces anti-cancer activity through oxidative stress-mediated DNA damage in circulating human lung cancer cells” (BCAM-D-18-01101R2). Those comments are of great value and help for revising and improving the quality of the paper. We have studied the comments carefully and made corrections. The main changes in the paper and the responses to the editor’s/reviewers’ comments are as follows:
Responses to the editor’s comments:

1. There are a number of statements and arguments that require further justification before they can be accepted.
   
   Reply: Thank you for your suggestion. We have discussed further the statements and arguments in the article.

2. There are some spelling and other grammatical errors which need to be modified.
   
   Reply: Thank you for your suggestion. We have modified the spelling and grammatical errors in the article.

Responses to the reviewers’ comments

(Reviewer 2):

1. The chemical constituents of the species contained in the product are not indicated, only fingerprint without information.
   
   Reply: Thank you for your advice. We also used traditional Chinese medicine standards for comparison, but due to different testing conditions, the relevant data has not shown.

   The results of the relevant standard comparison tests have been supplemented. For details, see supplementary material table s2.

2. You did not use any positive control in assay?
   
   Reply: Thank you for your suggestion. We did not have a positive control because there is currently no specific drug for CTC, but in future studies we will consider setting up a positive control.
3. Strengthen the discussion

Reply: Thank you for your suggestion. We have improved the discussion part.

We acknowledge the editor’s/reviewers’ comments and suggestions, which are valuable in improving the quality of our manuscript.

We tried our best to improve the manuscript and made some changes in the manuscript. These changes will not influence the content and framework of the paper.

We really appreciate Editor’s/Reviewers’ opinions, and hope that the correction will meet the approval criteria of the journal.

Once again, we would like to express our great appreciation to you and reviewers for comments on our paper.

Sincerely yours,

Dr. Jianhui Tian M.D., Ph.D.
Longhua Hospital Shanghai University of Traditional Chinese Medicine
Shanghai Institute of traditional Chinese Medicine Research Institute of TCM Oncology
725 Wanping Road, Shanghai 200032, People’s Republic of China
E-mail: hawk7150@hotmail.com